Author's response to reviews

Title: Developing and implementing an Accreditation system for Health Promoting Schools in Northern India: A cross-sectional study

Authors:

Jarnail S Thakur (jsthakur64@gmail.com)
Deepak Sharma (drdeepakpgimer@gmail.com)
Nidhi Jaswal (nidhi.jaswal@gmail.com)
Bhavneet Bharti (bavneetb@yahoo.com)
Ashoo Grover (ashoo groove@gmail.com)
Paramjyoti Thind (dr.paramjyoti@gmail.com)

Version: 2 Date: 8 March 2014

Author’s response to reviews: see over
Cover Letter

To,
The Editor,
BMC Public Health

Subject: - Submission of manuscript for publication in BMC Public Health

Sir/Madam,

Please find enclosed our manuscript entitled, ‘Developing and implementing an Accreditation system for Health Promoting Schools in Northern India: A cross-sectional study’ which we would like to submit for publication as a research article in journal, ‘BMC Public Health’. We confirm that this manuscript has not been published elsewhere and is not under consideration by another journal. All authors have approved the manuscript and agree with its submission to BMC Public Health.

The research area in which the authors have worked relates to a new area of school health and need of the time and this innovation needs to be in notice of the policy makers, health and education professionals as well as the general population as early as possible. All this could be possible if you kindly consider this research paper for publication. It is requested to waive off the publication fees (if any) because of low income country status of India and meager resources.

We have prepared this paper according to the norms of this journal. But if any alterations/editions are required, kindly intimate us or do it at your own end. Thanking you in anticipation. We look forward to hearing from you at the earliest convenience

Yours sincerely,

Corresponding Author,
Dr. J.S. Thakur, MD
Additional Professor,
School of Public Health,
Post Graduate Institute of Medical Education and Research
Chandigarh.
Mobile: 9463602173, Email Id: jsthakur64@gmail.com
Revisions in the manuscript

1. **Title Page:** Title Page has been added at the front of the manuscript file. It includes the authors' name, their affiliations, email addresses and full postal address of the corresponding author.

2. **Abstract:** The revised abstract as per the guidelines of biomedicalcentral has been uploaded.

3. **Ethics:** The details of ethics have been added in the manuscript.

4. **Authors’ contribution:** The revised format of authors' contribution as suggested has been added in the manuscript.